Author's response to reviews

Title: The Survey of the Health of Wisconsin (SHOW), a novel infrastructure for population health research: rationale and methods

Authors:

F. Javier Nieto (fjnieto@wisc.edu)
Paul E Peppard (ppeppard@wisc.edu)
Corinne D Engelman (cengelman@wisc.edu)
Jane A McElroy (mcelroyja@health.missouri.edu)
Loren W Galvao (lgalvao@uwm.edu)
Elliot M Friedman (friedman1@wisc.edu)
Andrew J Bersch (abersch@show.wisc.edu)
Kristen C Malecki (kmalecki@wisc.edu)

Version: 7 Date: 20 December 2010

Author's response to reviews:

We have corrected some typographical errors and did minor editing (mainly to eliminate a few rather verbose sentences and redundancies). NO substantial changes of any kind have been made.